Author’s response to reviews

Title: "You just need to leave the room when you breastfeed" Breastfeeding experiences among obese women in Sweden - A qualitative study

Authors:
Ing-Marie Claesson (ing-marie.claesson@liu.se)
Lotta Larsson (Lotta.E.Larsson@regionostergotland.se)
Linda Steen (linda7steen@gmail.com)
Siw Alehagen (siw.alehagen@liu.se)

Version: 2 Date: 19 Dec 2017

Author’s response to reviews:

Linköping December 18th 2017

Dear Editor

BMC Pregnancy and Childbirth,

Thank you for allowing us to revise our manuscript again. We have made the required editorial changes.

Sincerely

Ing-Marie Claesson,

Corresponding author on behalf of the other authors
1. In the Ethics and consent to participate section of the Declarations, please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. Written informed consent was obtained from all participators and this is stated in the section of the Declarations, page 23, lines 490 - 491 and in the manuscript, pages 6 - 7, lines 151 - 152.

2. Thank you for providing a consent to publish statement, however please revise this statement to clarify whether written consent to publish was obtained to publish the clinical information and the quotes in this manuscript. Written consent to publish clinical information and quotes in the manuscript was obtained from all participants. This is stated in the section of Declarations, page 23, lines 492 – 493.

If written consent was not obtained to publish participant information, please amend table 2 to show age-ranges instead of the ages of the participants and state “not applicable”.

3. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case. The statement is included in the section of Declarations, page 23, line 502.

4. Please note that figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. The Figure Legend is now placed after the References, page 26, lines 617 – 618. We have also removed the Figure title in the Figure file.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.